Author's response to reviews

Title: Epstein-Barr virus infection and clinical outcome in breast cancer patients correlate with immune cell TNF-alpha/IFN-gamma response

Authors:

Gina Marrao (gmarrao@gmail.com)
Mohammed Habib (mhabib@embl.fr)
Artur Paiva (Artur.Paiva@ipst.min-saude.pt)
Dominique Bicout (bicout@ill.fr)
Catherine Fallecker (cfalleck@embl.fr)
Sofia Franco (smp.franco@hotmail.com)
Samira Faffi-Kremer (samira.fafi-kremer@chru-strasbourg.fr)
Teresa Simões da Silva (teresassoliveira@gmail.com)
Patrice Morand (PMorand@chu-grenoble.fr)
Carlos Freire de Oliveira (cmdf.deoliveira@sapo.pt)
Emmanuel Drouet (drouet@embl.fr)

Version: 6 Date: 31 July 2014

Author's response to reviews:

Grenoble July 31st, 2014

To the Editor, BioMedCentral Cancer

Regarding your last comment

"EDITORS COMMENTS:

Regarding 3. Line 243 We put the sentence: "The PCR was consistently negative in the serum of all the BC patients. ", the sentence is missing.

Please upload a final version of article"

We apologize for that. We did check and confirm the sentence was put correctly at the line 243 ("Results" section, § "Global results and EBV status", 10th line from the beginning of the section: "The PCR was consistently negative in the serum of all the BC patients."

Yours sincerely,

Pr Emmanuel Drouet